Author's response to reviews

Title: Making sense of health information technology implementation: A qualitative study protocol

Authors:

Rebecca R Kitzmiller (rebecca.kitzmiller@duke.edu)
Ruth A Anderson (Ruth.anderson@duke.edu)
Reuben R McDaniel Jr (reuben.mcdaniel@mccombs.utexas.edu)

Version: 5 Date: 2 November 2010

Author's response to reviews:

Dear Implementation Science Editorial Team,

I have accepted all of the changes recommended by the editor. There were no specific queries contained within the manuscript which require a response from me.

I have uploaded the revised manuscript as well as revised appendices.

Best regards,

Rebecca Kitzmiller